Author’s response to reviews

Title: Hospital mortality in acute coronary syndrome: adjustment of GRACE score by D-dimer enables a more accurate prediction in a prospective cohort study

Authors:

Tongtong Yu (workhard31@vip.163.com)
Yundi Jiao (jiaoyundi1987@163.com)
Jia Song (836214497@qq.com)
Dongxu He (1570057021@qq.com)
Jiake Wu (wujiake1107@163.com)
Zhijun Sun (sunzj@sj-hospital.org)
Zhaoqing Sun (sunzhaoqing@vip.163.com)

Version: 2 Date: 23 Oct 2019

Author’s response to reviews:

Dear Editors and Reviewers:

Thank you for your letter and for the reviewers’ comments concerning our manuscript entitled "Hospital mortality in acute coronary syndrome: adjustment of GRACE score by D-dimer enables a more accurate prediction in a prospective cohort study" (BCAR-D-19-00134R1). Those comments are all valuable and very helpful for revising and improving our paper, as well as the important guiding significance to our researches. We have studied comments carefully and have made correction which we hope to meet with approval. The main corrections in the paper and the responds to the reviewer’s comments are as flowing:

Responds to the reviewer’s comments:

1. Response to comment (Reviewer 1): (sentence of acknowledgement confusing and ambiguous. Would suggest deleting acknowledgement since it seems that you have no special acknowledgement or just stated none in the acknowledgement section)

Response: Thank you for your comment. We have changed it.

2. Response to Editor Comments: (2. Overlap -- We note that the current submission contains some textual overlap with other previously published works)

Response: Thank you for your comment. We have rewritten the overlap.
3. Response to Editor Comments: (3. Availability of data and materials)
Response: Thank you for your comment. According to BioMed Central's editorial policies page, we have changed the statement.

4. Response to Editor Comments: (4. Additional files list-- Please add a section "Additional files")
Response: Thank you for your comment. We have added a section "Additional files".

5. Response to Editor Comments: (5. Clean manuscript)
Response: Thank you for your comment. We have uploaded our manuscript as a single, final, clean version.

We tried our best to improve the manuscript and made changes in the manuscript. These changes will not influence the content and framework of the paper. We appreciate for Editors/Reviewers’ warm work earnestly, and hope that the correction will meet with approval. Once again, thank you very much for your comments and suggestions.

Yours
Sincerely

Zhaoqing Sun